Author's response to reviews

Title: Testing the effectiveness of a mindfulness-based intervention to reduce emotional distress in outpatients with diabetes (DiaMind): design of a randomized controlled trial.

Authors:

Jenny van Son (J.vanSon1@uvt.nl)
Ivan Nyklicek (I.Nyklicek@uvt.nl)
Victor JM Pop (V.J.M.Pop@uvt.nl)
François Pouwer (F.Pouwer@uvt.nl)

Version: 5 Date: 9 February 2011

Author's response to reviews:

Dear Editor,

We checked the manuscript thoroughly, after which we made some small changes. These changes are indicated by means of 'track changes'. We apologize for the inconvenience.

On behalf of all authors,

Yours sincerely,

Jenny van Son MSc.